Author’s response to reviews

Title: Demographic and circumstantial accounts of burn mortality in Cape Town, South Africa, 2001-2004: An observational register based study

Authors:

A van Niekerk (Ashley.vanniekerk@mrc.ac.za)
R Laubscher (Ria.Laubscher@mrc.ac.za)
L Laflamme (lucie.laflamme@ki.se)

Version: 6 Date: 12 August 2009

Author’s response to reviews: see over
Dear Nina

Re:  MS:  3872691322537895
    Demographic and circumstantial accounts of fatal burn injuries in Cape Town. A
    register based cross-sectional study

Many thanks for facilitating this submission. We have made some minor adjustments to
Tables 2 to 4 to ensure that they are consistent with the changes made to Table 1, i.e. with
respect to horizontal lines, font size and spacing. Many thanks again for facilitating our
submission. Please contact me if you require any further adjustments.

Yours sincerely

Lucie Laflamme
Professor
Chair of the Department of Public Health Sciences Karolinska Institutet Division of
International Health Nobels väg 9 SE 171 77 Stockholm Sweden Tel +46 (0)8 524 83362
fax + 46 (0)8 311590
Department: www.phs.ki.se
research group: www.phs.ki.se/ihcar/research/isac